Author's response to reviews

Title: Massive Variceal Bleeding Secondary To Splenic Vein Occlusion Successfully Treated By Splenic Artery Embolization

Authors:

Theodossis S Papavramidis (papavramidis@hotmail.com)
Daniel Paramythiotis (danosprx@med.auth.gr)
Konstantinos Giavroglou (papavramidou@hotmail.com)
Stamati Potsi (matinapotsi@hotmail.com)
Fotis Girtovitis (pliiakos@hotmail.com)
Antonis Michalopoulos (amichal@med.auth.gr)
Vassilis N Papadopoulos (bnpap2003@yahoo.com)
John Prousalidis (vfduros@yahoo.gr)

Version: 2 Date: 6 January 2010

Author's response to reviews:

I want to thank all the reviewers for their very constructive comments. Especially, I want to thank Dr. Dandov for his spending time on correcting my article. All changes proposed by Dr. Dandov were welcome. All changes in the article are pinpointed by the track changes system of Word.